

## 

**Citation:** Fuentes-Hernandez A, Plucain J, Gori F, Pena-Miller R, Reding C, Jansen G, et al. (2015) Using a Sequential Regimen to Eliminate Bacteria at Sublethal Antibiotic Dosages. PLoS Biol 13(4): e1002104. doi:10.1371/journal.pbio.1002104

Academic Editor: Andrew Fraser Read, The Pennsylvania State University, UNITED STATES

Received: September 22, 2014

Accepted: February 11, 2015

Published: April 8, 2015

**Copyright:** © 2015 Fuentes-Hernandez et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Whole-genome sequence data with 18 samples and an annotated draft genome are available from the European Nucleotide Archive (ENA) with study accession number PRJEB7832. These data can be downloaded from http://www.ebi.ac.uk/ena/data/view/PRJEB7832. All other data are supplied as supplementary information.

Funding: RB was funded by grants EP/I00503X/1 and EP/I018263/1 from the UK Engineering and Physical Sciences Research Council. See: <u>http://gow.epsrc.ac.uk/NGBOViewPerson.aspx?PersonId=67077</u>. The funders had no role in study design, data RESEARCH ARTICLE

## Using a Sequential Regimen to Eliminate Bacteria at Sublethal Antibiotic Dosages

Ayari Fuentes-Hernandez<sup>1®</sup>, Jessica Plucain<sup>2®</sup>, Fabio Gori<sup>2®</sup>, Rafael Pena-Miller<sup>1</sup>, Carlos Reding<sup>2</sup>, Gunther Jansen<sup>3</sup>, Hinrich Schulenburg<sup>3</sup>, Ivana Gudelj<sup>2</sup>, Robert Beardmore<sup>2</sup>\*

1 Centro de Ciencias Genómicas, Universidad Nacional Autónoma de México, Cuernavaca, México, 2 Biosciences, Geoffrey Pope Building, University of Exeter, Exeter, United Kingdom, 3 Evolutionary Ecology and Genetics, Christian-Albrechts-Universität zu Kiel, Kiel, Germany

• These authors contributed equally to this work.

\* r.e.beardmore@exeter.ac.uk

## Abstract

We need to find ways of enhancing the potency of existing antibiotics, and, with this in mind, we begin with an unusual question: how low can antibiotic dosages be and yet bacterial clearance still be observed? Seeking to optimise the simultaneous use of two antibiotics, we use the minimal dose at which clearance is observed in an in vitro experimental model of antibiotic treatment as a criterion to distinguish the best and worst treatments of a bacterium, Escherichia coli. Our aim is to compare a combination treatment consisting of two synergistic antibiotics to so-called sequential treatments in which the choice of antibiotic to administer can change with each round of treatment. Using mathematical predictions validated by the E. coli treatment model, we show that clearance of the bacterium can be achieved using sequential treatments at antibiotic dosages so low that the equivalent twodrug combination treatments are ineffective. Seeking to treat the bacterium in testing circumstances, we purposefully study an E. coli strain that has a multidrug pump encoded in its chromosome that effluxes both antibiotics. Genomic amplifications that increase the number of pumps expressed per cell can cause the failure of high-dose combination treatments, yet, as we show, sequentially treated populations can still collapse. However, dual resistance due to the pump means that the antibiotics must be carefully deployed and not all sublethal sequential treatments succeed. A screen of 136 96-h-long sequential treatments determined five of these that could clear the bacterium at sublethal dosages in all replicate populations, even though none had done so by 24 h. These successes can be attributed to a collateral sensitivity whereby cross-resistance due to the duplicated pump proves insufficient to stop a reduction in E. coli growth rate following drug exchanges, a reduction that proves large enough for appropriately chosen drug switches to clear the bacterium.

collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

Abbreviations: DOX, doxycycline; ERY, erythromycin; IC<sub>x</sub>, inhibitory concentration (with an x % degree of inhibition); MIC, minimal inhibitory concentration; MRSA, methicillin-resistant *Staphylococcus aureus*; NCS, nonreciprocal collateral sensitivity; OD, optical density; *rrn*, ribosomal RNA operon; SNP, single nucleotide polymorphism.

#### Author Summary

So-called "cocktail" treatments are often proposed as a way of enhancing the potency of antibiotics, based on the idea that multiple drugs can synergise when used together as part of a single combined therapy. We investigated whether any other multidrug deployment strategies are as effective as—or perhaps even better than—synergistic antibiotic combinations at reducing bacterial densities. "Collateral sensitivities" between antibiotics are frequently observed; this is when measures taken by a bacterium to counter the presence of one antibiotic sensitise it to the subsequent use of another. Our approach was to see if we could exploit these sensitivities by first deploying one drug, then removing it and instead deploying another, and then repeating this process. This is not an entirely new idea, and there is a precedence for this form of treatment that has been trialled in the clinic for Helicobacter pylori infection. The idea we pursued here is an extension of "sequential treatment"; we investigated whether with two antibiotics and *n* rounds of treatment, if we search within the set of all possible  $2^n$  "sequential treatments"—including the two singledrug monotherapies-there might be treatments within that set that are more effective than the equivalent two-drug cocktail. Using a simple in vitro treatment model, we show that some sequential-in-time antibiotic treatments are successful under conditions that cause the failure of the cocktail treatment when implemented at the equivalent dosage.

#### Introduction

Bacteria have a remarkable capacity to adapt and evolve. It is probably unsurprising in retrospect that resistance has developed to every antibiotic in clinical use  $[\underline{1}]$ , with the genes responsible disseminated globally  $[\underline{2},\underline{3}]$ . Antibiotic resistance, therefore, has the potential to become a very grave problem. Bacteria evolve so rapidly, in fact, that whole-genome sequencing studies have been able to elucidate dozens of de novo drug-resistance mutations occurring at high frequency within a clinical patient's infection during a 12-wk treatment  $[\underline{4}]$ . Given this, the following seems an important question: what ways of combining antibiotics might be used to combat infection even when the bacterial species in question exhibits rapid decreases in drug susceptibility during treatment? Or, to put it differently, how can we enlarge the "optimisation space" of antibiotic combinations and search within those for novel, effective treatments?

One possibility may lie with so-called sequential treatments. They have been the subject of several recent laboratory studies [5-7] and clinical trials [8,9] in which the idea is to alternate the use of different antibiotic classes through time. Thus, if, for example, two antibiotics are available and *n* rounds of treatment are to be given, then there are  $2^n$  different ways of administering the drugs. Our hypothesis states that this exponentially large optimisation space can contain more effective treatments than the equivalent two-drug combination treatment when the same dosages of each antibiotic are applied.

We demonstrate the veracity of this claim in one particular in vitro laboratory model that mimics something of the gravity of the situation we now face by using a bacterium that possesses a scalable drug efflux mechanism that quickly reduces the efficacy of the antibiotics at our disposal. Despite this mechanism, we show that sequential treatments can clear the bacterium when the equivalent combination treatment fails to, provided, that is, that the drugs are deployed in a suitably optimised, sequential manner.

To demonstrate this, we use the following laboratory system. *Escherichia coli* K12 (AG100) is targeted with two antibiotics, erythromycin (a macrolide, ERY) and doxycycline (a tetracycline, DOX), that bind to different ribosomal RNA subunits, thereby inhibiting translation.

While this is a nonclinical drug pairing, the commercial drug Synercid (comprising quinupristin and dalfopristin) also targets ribosomal RNA combinatorially [10]. Moreover, some clinical combinations have ambiguous pharmacological interactions that can appear antagonistic in vitro [11,12], whereas the ERY—DOX pairing has an established synergy [13,14].

Before continuing, we need to declare a standard notational device that we will use throughout. It defines how antibiotic efficacy is measured, independently of the drug under study. Thus, IC<sub>x</sub> will denote the antibiotic concentration that reduces the density of the ancestral bacterial strain (AG100), rather than (for example) any other fitness measure, exponential growth rate, or area under a growth curve, by a factor x% relative to that produced without antibiotic in any single period of bacterial growth.

Now, *E. coli* is known to decrease susceptibility to ERY and DOX by amplifying a genomic region that contains the operon *acrRAB* because a multidrug pump is formed from the products of *acrRAB* and *tolC* [13,15]. Selection for amplification mutations occurs even when the drugs are combined at high concentrations whereupon pump duplications and triplications are observed [16]. The triplications permit bacteria subjected to 5 d of combination treatment at twice IC<sub>95</sub> dosages, and thus at very low population densities, to eventually restore their growth rates and population densities to almost untreated levels [16].

In these circumstances, the successful clearance of *E. coli* using sublethal dosages of ERY and DOX appears implausible. Low-dose monotherapies are unlikely to work [17], and combining the antibiotics into a synergistic IC<sub>50</sub> cocktail (that achieves IC<sub>90</sub> overall because of the synergy) is known to be futile because of resistance increases provided by the pump duplications [13]. We therefore turn to sequential treatments, an approach that has been used to treat cancers [18–20] and some clinical infections [9]. These might also appear predestined to fail; after all, cross drug collateral sensitivities are believed to be the basis of successful sequential treatments [7], whereas our model system, by contrast, has a scalable multidrug pump at its disposal.

Nevertheless, to evaluate the impact of extended antibiotic treatments, we propagated populations of *E. coli* in 96-well microtitre plates containing liquid medium supplemented with antibiotics based on 12-h cycles, aka seasons, of growth. Thus, two drug treatments per day were administered. At the end of each season, 1% of the spent liquid media, containing biomass, was transferred to a plate containing fresh medium and antibiotics, where growth could resume. The media was supplemented with enough glucose that this protocol would not clear the bacterium in the absence of drug but would instead establish a near-constant, season-by-season total observed population density of about  $10^8$  cells per ml in stationary phase (as can be discerned from Fig S1 and Fig S7 in S1 Text). Given this model, we sought antibiotic treatments capable of clearing the bacterium.

#### Results

## Low-Dose (IC<sup>50</sup>) and Mid-Dose (IC<sup>70</sup>) Sequential Drug Screens

By the term sequential treatment, we mean the following protocol: one of the two drugs is used in season 1, and, whether ERY or DOX, it may be re-used in season 2, or, alternatively, the other drug may be deployed instead. This process then continues each season until treatment ends. For a treatment of eight seasons, there are  $2^8-2 = 254$  possible sequential protocols (minus the two monotherapies). However, seeking to understand whether drug switches per se reduce population growth, only balanced sequential treatments that use four seasons of both drugs were trialled (Fig S6 in <u>S1 Text</u>, section 1). Seeking evidence of successful low-dose treatments, we first treated *E. coli* with ERY and DOX for eight seasons at dosages corresponding to the IC<sub>50</sub> of each drug, implementing the following treatments: two monotherapies, one 50/50 combination using a half dose of both drugs (achieving approximately  $IC_{90}$ , Fig S3 and Fig S4 in <u>S1 Text</u>, section 1) in addition to 70 sequential treatments (three replicates each). An analogous screen of sequential treatments was then implemented at  $IC_{70}$  dosages (but only 66 of these sequential treatments were implemented).

Fig. 1 summarises the IC<sub>50</sub> data. In Fig. 1A, the 50/50 combination treatment achieves greater single-season inhibition than each monotherapy, as expected from prior reports of synergy  $(p<10^{-7}, \text{test} \text{ as indicated in Fig S3 in S1 Text})$ . However, by 36 h the combination therapy no longer produces the lowest bacterial densities, and by 96 h it produces high final densities (Fig. 1A and Fig. 1B), higher than the mean of the family of sequential treatments ( $p<10^{-8}$ , F (1,69)≈47.1, one-way ANOVA). Although a sequential treatment has the lowest final density of all those trialled (Fig. 1A), no IC<sub>50</sub> treatment provided any evidence of eliminating the bacteria by 96 h.

After increasing dosages to their  $IC_{70}$  values, the following evidence of bacterial clearance by 96 h was observed. Sixteen sequential treatments that produced some of the lowest population densities after 96 h of treatment (treatments marked with boxes in Fig. 2A) were examined, and, using spot tests, we could isolate no live cells for five of these treatments in all three replicates. The 11 remaining treatments lead to a zero cell count in some replicates but not in all (Fig S15 in S1 Text, section 3). We then replicated all 16 treatments an additional three times, and the five previously successful treatments again produced a zero cell count by 96 h, although the remaining 11 treatments showed substantial between-replicate variability in their population dynamics (Fig S15). By contrast, Fig. 2B shows that the 50/50 combination treatment (with a greater inhibition than  $IC_{70}$  due to the synergy) and both monotherapies yielded recovering (i.e., increasing) mean population densities beyond 48 h at these dosages. (In addition, we recall that twice  $IC_{95}$  combinations of these drugs can fail in this treatment model too [16].) However, these observations serve to illustrate that appropriately optimised, sequential therapies at  $IC_{70}$  can clear a bacterium even when synergistic combination treatments with greater one-season inhibition do not.

In order to determine genetic changes due to the differential stresses found in drug-free conditions and in the sequential and combination treatments, two treatments at IC<sub>50</sub> that produced comparable densities at 96 h were subjected to a whole-genome sequencing analysis and compared to the drug-free populations (<u>S1 Text</u>, section 4). Writing "E" for a season of ERY and "D" for DOX, when metagenomes from the EDEDEDED and 50/50 combination treatments were sequenced, known resistance mutations were observed in both. Fig. <u>3B</u> highlights a 412 Kb genomic region containing the *acrRAB* operon whose duplication was observed more frequently in both the combination and sequential treatments at 96 h (namely, eight seasons) than at 24 h (or two seasons; Fisher exact test for both, p = 0.05; Fig S17 in <u>S1 Text</u>, section 4). Treating sequentially does not, therefore, avert selection for duplications of the *acrRAB* operon.

We sought evidence for triplications of *acrRAB* by asking whether the ratio of coverage depths between amplified and nonamplified genomic regions was above a value of 2, the latter being the maximum value possible of this statistic if no triplications were present. However, at 96 h, in neither the sequential treatment (one-sided *t* test,  $p\approx0.12$ ,  $T\approx1.68$ , n = 3) nor the 50/50 combination treatment (one-sided t-test,  $p\approx0.061$ ,  $T\approx2.60$ , n = 3) was this value significantly above 2. Finally, single nucleotide polymorphisms (SNPs) were observed in the putative drug transporter gene *mdtG* (*yceE* [21]) in all conditions (Table S3 in S1 Text, section 4); this member of the *marA-soxS-rob* stress regulon mediates expression of the *acrAB-tolC* pump [22].

We expected the rate of adaption (defined as a rate of increase in growth rate [14]) to correlate positively with dose. Instead, we observed that adaptation can be just as rapid in the absence as in the presence of antibiotics (Fig. 3A), and our culture conditions may explain this.



Fig 1. At IC<sub>50</sub> dosages, population recovery is fastest for the 50/50 combination treatment and slowest for a sequential treatment. (A) Mean densities are shown at the end of each season for all sequential treatments at IC<sub>50</sub> (as blue and green dots) and for the 50/50 combination of both drugs (black dotted line). The treatment maximising inhibition in season 1 (at 12 h) is the 50/50 combination treatment, because of the synergy. However, by season 8 (at 96 h), all sequential treatments produce lower mean densities than the 50/ 50 treatment, out of which the lowest density obtained from all the treatments tested is indicated by red circles. Also shown are mean final densities (see x-label "means") of the 50/50 treatment (black circle), the best sequential treatment (red circle), and of all sequential treatments (green circle ± SE, three replicates per treatment). (B) A forest plot showing densities obtained using different sequential treatments at 96 h relative to the 50/50 treatment (drug orders are illustrated by the blue and green boxes on the left). The vertical black line represents the mean density for the 50/50 combination, the vertical dashed line is the mean of all the sequential treatments, and the dots mark the deviation in density produced from the 50/50 combination treatment ( $\pm$  SE, n = 3). Like (A), this shows that the combination treatment performs at the poorest extreme of the distribution of all sequential treatments measured in terms of how bacterial growth is suppressed by 96 h. There is no evidence of bacterial clearance in any treatment. (S1 Data contains the data used in this figure.)

Slow growing cells, like persister phenotypes [23] and small colony variants [24], are cleared by our protocol, whereas cells that achieve rapid growth above approximately 6.6 generations every 12 h can survive. Rapid bacterial growth is associated with physiological changes that include negative DNA supercoiling and multiple DNA replication forks per cell [25], increased cell size [26], and heightened ribosomal demand [27]. The latter likely induced a stringent response in the fastest growth conditions (the absence of drug). In these conditions, SNPs associated with fatty acid degradation, lipid peroxidation stress, and sulphur transportation (*tauA*) were observed, the latter at high frequency (Table S3 in S1 Text, section 4). As tauA is expressed in our growth media only during cysteine limitation [28], overcoming  $\alpha$ -amino acid starvation is a likely mechanism supporting the SNPs detected in all seven 23S ribosomal RNA operons (rrn) of E. coli by 96 h in the absence of drugs (Table S1 in S1 Text, section 4). Although mutations in the same rrn loci were observed at low frequency at 96 h in the slowergrowing populations treated sequentially with drugs, none of these operons were mutated in populations treated with the drug combination. We hypothesise, therefore, that the antibiotics have slowed the rise and sweep of adaptive mutations needed for optimal growth in our culture conditions (Table S1 in S1 Text, section 4). Finally, we found no significant evidence of SNPs within drug targets in any conditions (S1 Text, section 4).



**Fig 2.** Some examples of successful sequential treatments at IC<sub>70</sub> dosages. (A) This Manhattan plot at IC<sub>70</sub> shows the mean total optical densities observed during eight seasons of treatment ( $\Sigma OD(T)$  on the vertical axis, vertical lines are SE, and n = 3). Note the 16 treatments marked with a red or black square: they had among the lowest final densities of the treatments trialled. After this, the 16 treatments were replicated; a red square shows that a zero cell density was observed in all three initial and subsequent replicates of that treatment, and black squares show a zero population density was observed in some, but not all, replicates. (B) The no-drug, ERY, and DOX monotherapies and the 50/50 combination treatment all produce recovering mean population densities at IC<sub>70</sub> doses. These four unsuccessful treatments are shown next to the optical density dynamics of three replicates of a successful "red square" treatment from (A) (treatment C in panel C). The three replicates (shown as grey lines with blue [DOX] and green [ERY] circles) indicate parallel dynamics and fluctuating decay towards zero (bars are SE of optical density at 96 h, n = 3). (C) Season-by-season mean densities of all the successful (red square) treatments from (A); note how two achieve high densities early during treatment. Fig S15 in <u>S1 Text</u>, section 3, shows colony-forming units for replicates of these treatments. (S1 Data contains the data used in this figure.)

#### Nonreciprocal Collateral Sensitivity

Antibiotic combinations are used to slow drug-resistance adaptation because they enhance their antibacterial effect through inhibitory synergisms [29] and because they reduce the number of potential resistance mutations. Here, consistent with prior studies of ERY—DOX combinations [13], growth rate adaptation is so rapid when using ERY and DOX in a synergistic IC<sub>70</sub> combination that the bacterium is not cleared (Fig. 1A, Fig. 2B, Fig. 3A), and an analogous observation has already been made at double IC<sub>95</sub> dosages [16].

seasons

Collateral sensitivities on the other hand, in which the prior use of one antibiotic sensitises the bacterium to the use of another, have few recognised mechanisms [30], but they too have been proposed as a possible basis for successful sequential treatments [7,31] because the change

seasons



**Fig 3. The rate of adaptation has a complex relationship with antibiotic dose.** (A) Using IC<sub>50</sub> dosages, growth rate adaptation (denoted  $\alpha$  when defined in [14]) is greater for the 50/50 combination than for the EDEDEDED ("ERY—DOX") and DEDEDEDE ("DOX—ERY") sequential treatments. Adaptation can also be faster in the absence, rather than in the presence, of antibiotics. The right-hand plot shows each replicate separately (as a dot), indicating treatment clusters as coloured regions using the convex hulls of the datasets for each treatment type (whether no drug or single drug monotherapies, the 50/50 two-drug combination, or a sequential treatment). This shows sequential treatments minimise both final growth rate after eight seasons and the rate of adaptation. (B) Differences in the rate of adaptation in (A) are not accounted for by the duplication of the *acrRAB* operon, and sequential treatments do not prevent pump duplications. These coverage plots from sequenced populations at 24 h and 96 h show that both the combination (50/50) and sequential treatments (ERY/DOX) lead to the duplication of a genomic region from 273 Kb to 686 Kb that contains *acrRAB*. Left: the duplication was absent from all treatments after 24 h. Right: the duplication (the dark sector) is present in both the 50/50 combination and ERY—DOX sequential treatments after 96 h, but not in the no-drug control. Single nucleotide polymorphisms (SNPs) are highlighted as arrowheads next to the treatment in which they were observed. (S1 Data contains the data used in this figure.)

PLOS BIOLOGY

of environment hampers adaptation. Promisingly, the rate of adaptation is demonstrably lower here for sequential treatments than for combinations (Fig. 3A;  $p < 10^{-4}$ ,  $F(2,21) \approx 16.8$ , one-way ANOVA with Bonferroni correction). However, it has also been suggested that the antibiotic sequences should follow optimised pathways through networks of drug choices that maximally

sensitise the bacterium to treatment [ $\mathbb{Z}$ ]. However, *E. coli* AG100 has pumps for both ERY and DOX, and we might therefore expect to observe cross-resistance, not collateral sensitivity, for this drug pair and this bacterium [ $\underline{6}$ ]. It therefore appears we do not have enough drugs for drug cycling to work in this model, but it will turn out, in fact, that we do.

This is because at least two cross sensitivity properties are observed. The first of these, which was noted recently for doxycycline [6], we term *nonreciprocal collateral sensitivity* (NCS), and it is defined as follows. Label two drugs "A" and "B" and choose equivalent dosages for both, meaning IC<sub>x</sub> for some x, and let  $D(T_1,T_2)$  denote density of the population when treatment  $T_2$  follows treatment  $T_1$ . We will also use the notation  $A^n$  and  $B^n$  to denote monotherapies with n rounds of treatment. Now, suppose we begin with a clonal population and treat with A for n+1 time units so that  $D(A^n,A)$  denotes the population density after the (n+1)-th treatment. Then, in a separate experiment, we treat with A for n time units followed by B for one time unit so that  $D(A^n,B)$  denotes the final population density. A nonreciprocal collateral sensitivity between A and B is said to occur when the switch from A to B results in a density decrease so that  $D(A^n,B) < D(A^n,A)$ , whereas an analogous switch from B to A results in a density increase, meaning  $D(B^n,A) > D(B^n,B)$ . When satisfied, this definition means A-adapted populations appear sensitised to drug B, whereas B-adapted populations have increased resistance to A.

If present, an NCS demonstrates that single-season inhibitory values cannot be used to infer the later inhibitory effect of antibiotics as the treatment proceeds; thus,  $IC_X$  values and rate of adaptation measures capture very different properties of the bacterium. For example, despite both drugs having equivalent inhibitory effects on a wild-type population after one dose, drugresistance mutations could sweep more rapidly for one drug than the other, and this could result in an NCS. Nevertheless, if the observed collateral sensitivity is much larger than the crossresistance within a dataset that indicates the presence of an NCS, appropriately chosen sequential regimens may still be sufficiently potent to eliminate the bacterium.

# Postulating a Mechanism That Supports Nonreciprocal Collateral Sensitivity

We first sought collateral sensitivities within the entire dataset shown in Fig. 2A but found no significant evidence (Fig S12 in S1 Text, section 3) that a switch from ERY to DOX had a different effect on population density than switching from DOX to ERY. We therefore tested for the presence of an NCS using a simpler "(n+1)-protocol": *n* seasons of culture with one drug, followed by a switch to the other drug for just one season's duration. This protocol (Fig. 4) shows that when AG100 is treated with ERY for *n* seasons (of 24 h duration) and DOX on the (n+1)-th season, both at IC<sub>70</sub>, the continued increase in bacterial density on the last season is consistent with cross-resistance (see Fig. 4B for *p*-values). However, when treating with DOX for *n* seasons and then ERY on the (n+1)-th, a density reduction is observed on the last treatment, consistent with a collateral sensitivity (Fig. 4B). This drug pair therefore possesses an NCS: although both inhibit growth of wild-type *E. coli* equally, they report different levels of inhibition on drug-adapted populations.



**Fig 4.** A nonreciprocal collateral sensitivity at  $IC_{70}$  dosages with respect to population densities. (A) The concentration of each drug was calibrated to ensure  $IC_{70}$  was achieved for both drugs, here at 18 h. (B) A nonreciprocal collateral sensitivity (NCS) determined using the (n+1)-protocol described in the text, where n =3,4,...,7: the change in densities following a change in antibiotic demonstrates ERY  $\rightarrow$  DOX cross-resistance but DOX  $\rightarrow$  ERY cross sensitivity (*p*-values from *t* tests, n = 5). (S1 Data contains the data used in this figure.)

This observation accords with predictions of the following theoretical model [13]:

$$\frac{d}{dt}b_1 = G(S, D_1, E_1)b_1 - \delta(b_1 - (1 + \Delta)b_2), \tag{1}$$

$$\frac{d}{dt}b_j = G(S, D_j, E_j)b_j - \delta((2+\Delta)b_j - b_{j-1} - (1+\Delta)b_{j+1}),$$
(2)

$$\frac{d}{dt}b_{N} = G(S, D_{N}, E_{N})b_{N} - \delta((1+\Delta)b_{N} - b_{N-1}),$$
(3)

$$\frac{d}{dt}S = -\frac{VS}{K+S}\sum_{j=1}^{N}b_j,\tag{4}$$

$$\frac{d}{dt}D_{ext} = -d_D D_{ext} - \sum_{j=1}^N b_j \left(\varphi_d (D_{ext} - D_j) - \frac{\nu_d p_j}{k_d + p_j} D_j\right),\tag{5}$$

$$\frac{d}{dt}D_j = -d_D D_j + b_j \left(\varphi_d (D_{ext} - D_j) - \frac{\nu_d p_j}{k_d + p_j} D_j\right),\tag{6}$$

$$\frac{d}{dt}E_{ext} = -d_E E_{ext} - \sum_{j=1}^N b_j \left(\varphi_e(E_{ext} - E_j) - \frac{\nu_e p_j}{k_e + p_j}E_j\right),\tag{7}$$

$$\frac{d}{dt}E_j = -d_E E_j + b_j \left(\varphi_e(E_{ext} - E_j) - \frac{\nu_e p_j}{k_e + p_j}E_j\right),\tag{8}$$

where j = 2, ..., N-1 is a parameter that controls the number of efflux genes each cell can

express. Equations <u>1–8</u> capture the densities,  $b_j$ , of bacteria with duplications of a gene that exports drugs from within the cell. At time t, S is the concentration of a limiting carbon source,  $D_{ext}$  and  $E_{ext}$  are extracellular concentrations of each drug, DOX and ERY respectively,  $D_j$  and  $E_j$  are the intracellular drug concentrations, and drugs degrade at rates  $d_D$  and  $d_E$ . The variable  $p_j$  represents the expected number of efflux pumps expressed by a cell with j-1 efflux genes, for each  $j\geq 2$ . We also assume that  $p_1 = 0$  so that it is possible for cells to encode the pump without expressing it (meaning genotypes for which j = 1 have the efflux gene but do not express it). More generally, the nonzero quantity  $p_{j+1}$  is defined, for  $j\geq 1$ , by  $j/(1+\gamma \cdot j)$ ), and then  $p_j/(k_e+p_j)$  is the probability that a given drug molecule is bound to an efflux pump. This model is a simplification of the competition each transcription unit has for each efflux operon (the *acrRAB* promoter), whereby a diminishing return is present in the number of pumps expressed as the number of efflux genes increases; the rate of diminishing returns is controlled by  $\gamma>0$ . Increases and decreases in the number of efflux genes in a cell are assumed to be a Poisson process with parameter  $\delta$  per cell per hour.

Other variables in Equations 1-8 have the following meaning:  $\varphi_e, \varphi_d$  are antibiotic diffusion rates across the cell membrane,  $v_e, v_d$  are maximal drug efflux rates, and  $k_e, k_d$  are half-saturation constants associated with pump-antibiotic binding; *V* and *K* are maximal uptake rate and half-saturation constants associated with a Michaelis-Menten uptake model of the limiting carbon source (*S*); growth rate  $G(S,D,E) = cVS/((1+k_eE+\kappa_dD+\kappa_{ed}ED)(K+S))$  is proportional to uptake rate via a per-sugar biomass yield constant *c*, and G(S, D, E) is reduced in value synergistically by the drugs (where  $\kappa_e, \kappa_d$  and  $\kappa_{ed}$  are parameters that control drug efficacy and strength of synergy);  $\delta$  is the rate of amplification of the efflux gene, and  $\delta(1+\Delta)$ , a value necessarily greater than  $\delta$ , is the rate of loss of the gene. *N*-1 is the maximum number of copies of the efflux gene. We set N = 3 to represent three different cell phenotypes: an unexpressed pump gene (a wild type), a single expressed pump gene, and one additional copy of that gene in which both copies are expressed. Finally, the model is simulated with several seasons so as to mimic the in vitro protocol, with the loss of 99% of all cells implemented at the end of each season. Equations 1-8 were solved numerically using a parameterisation determined from a prior training dataset (S1 Text, section 5) [13].

Although our theory does not capture all aspects of our data, computations show that, like *E. coli*, the model possesses an NCS (Fig. 5). The model predicts that a pump asymmetry due to different efflux efficiencies of ERY and DOX produces populations with differential susceptibility to each drug resulting from having different frequencies of drug-susceptible wild-type cells existing in mutation-selection equilibrium with less susceptible mutants (Fig. 5A). Supporting the hypothesis of different efflux efficiencies of ERY and DOX, data from the *E. coli acr* efflux knockout strain AG100A( $\Delta acr$ ) (Table S2) [13] shows that the loss of *acrB* reduces the IC<sub>50</sub> of ERY to approximately 5% of the wild-type AG100 value but reduces it to just 23% in the case of DOX. The model captures others features of the data, particularly that appropriately chosen sequential treatments produce fewer bacteria than the combination, yet some sequential treatments produce more (Fig S18 in S1 Text, section 5).

A second cross sensitivity property of the ERY—DOX was also established, as follows. Having found a mechanism for an NCS with respect to population densities, we hypothesised that the (n+1)-protocol data could exhibit cross sensitivities with respect to other measures of bacterial fitness. To demonstrate this, we fitted the logistic growth model  $\frac{dx}{dt} = R \cdot x(1 - x/K)$  to bacterial density time series, where the parameter *R* is per hour per capita growth rate and *K* is the population carrying capacity. The resulting data exhibits collateral sensitivities irrespective of the order in which the drugs were exchanged because a reduction of *R* was observed



**Fig 5.** A mathematical model indicates frequency-dependent selection for the pump duplication can cause a nonreciprocal collateral sensitivity profile with respect to population densities. (A) The first two columns indicate modelled internal drug concentrations in three cell phenotypes (dark blue: "wild-type" cells not expressing the pump; red: the pump gene is expressed; dark grey: two pump genes are expressed). The third column shows modelled population densities through time, indicating the frequencies of each phenotype within that density. The different drugs select for resistant (pump-expressing) and susceptible (pump-not-expressed) phenotypes at different rates, despite both having been calibrated to equal inhibitory effect (namely  $IC_{50}$ ) on a population consisting almost exclusively of wild-type cells by the end of day 1. These simulations show that monotherapies consisting of either drug select for different frequencies of the pump gene and its duplication, depending on which drug is being applied. Thus, given n days of adaptation to DOX followed by adaptation to ERY, after the switch, density decreases. (B) Implementing the (*n*+1) protocol in the model is consistent with the data of Fig. 4. (S1 Data contains the data used in this figure.)

following a change of drug for every *n* tested (from 3 to 6), although not all reductions were significant (Fig. 6).

#### Discussion

Despite the ability of bacteria to adapt to an antibiotic challenge, our laboratory model shows that one can exploit a sensitising property of fluctuating environments to eliminate a bacterium eventually at dosages that only inhibit growth by 70% initially. However, ours is a very simple treatment model inspired by bacterial infections, and we do not wish to overstate its predictive power in relation to the treatment of humans. In particular, the loss of slow-growing cells from our microcosm is not representative of the in vivo conditions in which slow-growing, antibiot-ic-tolerant phenotypes can be responsible for recalcitrance to treatment [32]. Our model is also limited because it lacks an immune response or any of the environmental complexity found in the human body.

The clinical practise of how antibiotics are used to treat bacterial infections begins with the minimal inhibitory concentration (MIC), the minimal drug concentration at which no visible growth of a bacterium is observed after overnight culture in vitro [33,34]. After determining the MIC, antibiotics are deployed at high enough dosages so that peak concentrations are achieved in vivo well in excess of this number [35]. Experience has shown that super-MIC dosages are necessary for successful recovery from infection when using combination treatment and monotherapy [33-39], although there is recent evidence of effective lower-dose antifungal, anti-MRSA (methicillin-resistant *Staphylococcus aureus*), and antimalarial treatments in vivo [40-42]. However, so few sequential treatments have been trialled in the clinic that there is little accumulated data on or understanding of what dosing or scheduling criteria might be needed when using sequential treatments in vivo.

Furthermore, not all  $IC_{70}$  sequential treatments lead to clearance, and some drug sequences produced higher population densities than the equivalent combination treatment (Fig. 2A).



**Fig 6. A collateral sensitivity at IC**<sub>70</sub> with respect to per capita growth rate, **R**. (A) Optical density time series data used in Fig. 4 was reused by fitting a logistic growth model (defined in the text) to estimate growth rates. For clarity, both the growth rate parameter *R* and the regression coefficient  $R^2$  from exemplar fits are indicated alongside modelled dynamics. (B) The resulting dataset shows significant and nonsignificant collateral sensitivities with respect to growth rate (*R*) following an exchange of antibiotic (*t* tests, *n* = 5). (S1 Data contains the data used in this figure.)

The reasons for this are not clear, although both data and the theoretical efflux model do exhibit large between-treatment variation (Fig. 2A, Fig S18 in S1 Text, section 5). A prior hypothesis states that greater antibiotic heterogeneities in the bacterial environment should diminish the rate of drug-resistance adaptation [43]. This would mean, for example, that bacteria adapt more readily to an EEEEDDDD treatment (one drug switch) than to EDEDEDED (seven switches). We therefore sought a relationship between bacterial population densities and the number of drug switches implemented in different sequential treatments but found no evidence of the predicted correlation (Fig S11 in S1 Text, section 3).

It is well known that the use of low dosages can select for resistant strains when they are competed in co-culture with susceptible strains [17]. However, that mid-dose clearance is still possible has a simple explanation in principle: the antibiotic sequences eventually reduce, and then maintain, Malthusian fitness of the evolving population below zero. To examine this in the simplest of theoretical contexts, suppose  $B_n$  represents bacterial population density after n rounds of treatment where 1-I, with 0 < I < 1, is an expected fraction of cells that are not cleared by the treatment but instead lost because of other effects (for example, host immunity) each

treatment. Indeed, granulocyte-mediated clearance has been shown to achieve a two-log<sub>10</sub> reduction in bacterial load in a 24 h period using a murine model [44], giving a value of  $I \approx 1/100$ .

At any sublethal dosage, whereby exponential population growth (at rate r) occurs between treatments spaced T time units apart (cf. Fig. 1A), it follows for a bacteriostatic antibiotic at a dose of "A units" that  $B_{n+1} = I \cdot B_n e^{(r-A)T}$ . Bacterial clearance is assured when the population decays eventually; this happens when  $B_{n+1}/B_n < 1$ , which is equivalent to the condition A > r-(log  $(\Gamma^1))/T$ . Note that this value is less than r by an amount that depends on I. If we now define an analogy of the MIC as, say, the IC<sub>99</sub> in this toy model, which is the antibiotic dose that reduces bacterial growth by 99% after one treatment, the condition on A to achieve IC<sub>99</sub> is  $e^{(r-A)T} < 1/$ 100, or  $A > r + (\log 100)/T$ . This value is greater than r and therefore effective therapeutically, but it is not representative of the critical minimal dose needed to clear the bacterium [45].

The absence of visual eradication overnight in vitro should not, according to this argument, itself be used as a rationale to preclude the practical use of an antibiotic drug. Indeed, the antifungal azole drugs are used to treat *Candida albicans* clinically at dosages that do not always eliminate population growth after overnight culture in vitro [45-47]. In our in vitro study, we needed to keep the rate of adaptation low for the above theoretical rationale to work, and only certain sequential treatments were able to do this (Fig. 3A and 3B). The requirement for low rates of adaptation likely needs the mutations and physiological changes that arise early during treatment, when population sizes are large, to provide no benefit, or even be deleterious later during treatment and so prevent recovery when population size, and mutational supply, is small.

There have been clinical successes for one particular sequential treatment: *Helicobacter pylori* infection has improved eradication rates for a sequential treatment relative to a combination therapy at the same dose [8,9], although geographical variations in successes have been observed and attributed to pathogen strain differences [48]. We hypothesise that the treatment of other clinical pathogens may be possible using sequential antibiotic treatments.

We note that low dosing is used to treat some bacterial infections. The ability of antibiotics to act as modulators of gene expression at low doses [49] can be exploited, for example when certain drug classes (including macrolides) are used to control the expression of virulence factors in MRSA [50]. Moreover, the use of  $\beta$ -lactam antibiotics as a low-dose adjuvant is a novel strategy in the treatment of recalcitrant MRSA infection [51,52], even though MRSA is resistant to most of these drugs. The  $\beta$ -lactam does not target the cell directly; rather, it enhances the activity of host peptides that are not antimicrobial per se but which modulate the host immune response [53].

However, it is not our intention to advocate for the indiscriminate clinical use of low-dose regimens. Rather, we are claiming that sequential dosing strategies exist for administering antibiotics that are sufficiently potent, and which prevent adaptation enough, to clear a bacterium when the equivalent dose combination treatment fails to do so. That this can be done even though the bacterium has a scaleable multidrug resistance mechanism in its chromosome gives us cause to hypothesise that new ways of optimising antibiotic use in vivo can be found by alternating them as part of treatment.

#### **Materials and Methods**

#### Media and Strains

We used *E. coli* AG100 (a gift from Stuart B. Levy) and M9 minimal media (0.2% glucose and 0.1% casamino acids). Stock solutions of DOX and ERY were made from powder stocks (Sigma-Aldrich) at 5 mg/ml in water for DOX and 100 mg/ml in ethanol for ERY and stored at -20°C. All subsequent dilutions were made from these stocks and kept at 4°C.

#### **Batch Transfer Protocol**

A microtitre plate reader measured optical densities every 20 min at 600nm as a proxy for population densities in different environments ( $R^2$ >0.99, Fig S1 in <u>S1 Text</u>, section 1). 96-well plates containing 150 µLof liquid per well incubated at 30°C were used to culture bacteria; these were shaken in a linear manner before each measurement was taken.

For prolonged exposure to antibiotics, inoculating bacteria were taken from one colony and cultured overnight in M9 minimal media (0.2% glucose, 0.1% casamino acids) at 30°C in a shaker-incubator. At the end of each season, a 96-pin replicator sampled the liquid volume, which was then transferred to a new plate containing fresh growth medium and antibiotics; the same environment for each replicate population was maintained. Every subsequent transfer was performed using the 96-pin replicator; the volume transferred was approximately  $1.5\mu$ L. OD time series were imported into Matlab R2013b to subtract the background (blank wells containing only medium) and generate all other statistics.

### Live Cell Counts

No claim is made on the basis of optical density data alone that a zero population density had resulted from treatment. Zero densities were determined by observing an OD value below  $10^{-2}$  units, whereafter the presence of cells was determined by spot tests. Serial dilutions were then used to determine live cell numbers in colony-forming units, if any were detected (Fig S15 in S1 Text, section 3).

#### WGS Data Accession Number

Whole-genome sequence data with 18 samples and an annotated draft genome is available from the European Nucleotide Archive (ENA) with study accession number PRJEB7832. This data can be downloaded from <u>http://www.ebi.ac.uk/ena/data/view/PRJEB7832</u>.

## **Supporting Information**

**S1 Data.** This file contains the data used to produce all figures in the main text. (XLSX)

**S2** Data. This file contains the data used to produce all figures in <u>S1 Text</u>. (XLSX)

**S1 Text. Supplemental information.** Contains experimental materials and methods, typical growth data and the rate of adaptation, additional data, a whole-genome sequencing analysis, an additional simulation of the mathematical model, and Figs. S1–S18, Tables S1–S6, and their references.

(PDF)

#### Acknowledgments

The authors thank Stuart Levy and Laura McMurry for providing bacterial strains and Konrad Paszkiewicz and Karen Moore for computational and laboratory support.

#### **Author Contributions**

Conceived and designed the experiments: RB HS GJ RPM AFH JP IG. Performed the experiments: AFH JP CR GJ. Analyzed the data: AFH CR FG RPM RB. Contributed reagents/materials/analysis tools: AFH HS IG RPM FG GJ RB. Wrote the paper: RB.

#### References

- Payne DJ, Gwynn MN, Holmes DJ, Pompliano L (2007) Drugs for bad bugs: confronting the challenges of antibacterial discovery. Nature Reviews Drug Discovery 6: 29–40. PMID: <u>17159923</u>
- Davies J, Davies D (2010) Origins and evolution of antibiotic resistance. Microbiol Mol Biol Rev 74: 417–433. doi: <u>10.1128/MMBR.00016-10</u> PMID: <u>20805405</u>
- Davies J (1994) Inactivation of antibiotics and the dissemination of resistance genes. Science 264: 375–382. PMID: <u>8153624</u>
- Mwangi MM, Wu SW, Zhou Y, Sieradzki K, de Lencastre H, et al. (2007) Tracking the *in vivo* evolution of multidrug resistance in *Staphylococcus aureus* by whole-genome sequencing. PNAS 104: 9451– 9456. PMID: <u>17517606</u>
- Kim S, Lieberman TD, Kishony R (2014) Alternating antibiotic treatments constrain evolutionary paths to multidrug resistance. Proceedings of the National Academy of Sciences 111: 14494–14499. doi: <u>10.</u> <u>1073/pnas.1409800111</u> PMID: <u>25246554</u>
- Oz T, Guvenek A, Yildiz S, Karaboga E, Tamer YT, et al. (2014) Strength of selection pressure is an important parameter contributing to the complexity of antibiotic resistance evolution. Molecular Biology and Evolution 31: 2387–2401. doi: 10.1093/molbev/msu191 PMID: 24962091
- 7. Imamovic L, Sommer M (2013) Use of collateral sensitivity networks to design drug cycling protocols that avoid resistance development. Science Translational Medicine 5: 204ra132.
- Kim YS, Kim SJ, Yoon JH, Suk KT, Kim JB, et al. (2011) Randomised clinical trial: the efficacy of a 10day sequential therapy vs. a 14-day standard proton pump inhibitor-based triple therapy for *Helicobacter pylori* in Korea. Aliment Pharmacol Ther 34: 1098–1105. doi: <u>10.1111/j.1365-2036.2011.04843.x</u> PMID: 21923713
- Vaira D, Zullo A, Hassan C, Fiorini G, Vakil N (2009) Sequential therapy for *Helicobacter pylori* eradication: The time is now! Therap Adv Gastroenterol 2: 317–322. doi: <u>10.1177/1756283X09343326</u> PMID: <u>21180579</u>
- Fischbach MA (2011) Combination therapies for combating antimicrobial resistance. Curr Opin Microbiol 14: 519–523. doi: <u>10.1016/j.mib.2011.08.003</u> PMID: <u>21900036</u>
- Bayer AS, Morrison JO (1984) Disparity between timed-kill and checkerboard methods for determination of *in vitro* bactericidal interactions of vancomycin plus rifampin versus methicillin-susceptible andresistant Staphylococcus aureus. Antimicrob Agents Chemother 26: 220–223. PMID: 6567464
- Deresinski S (2009) Vancomycin in combination with other antibiotics for the treatment of serious methicillin-resistant *Staphylococcus aureus* infections. Clin Infect Dis 49: 1072–1079. doi: <u>10.1086/605572</u> PMID: <u>19725789</u>
- Pena-Miller R, Laehnemann D, Jansen G, Fuentes-Hernandez A, Rosenstiel P, et al. (2013) When the most potent combination of antibiotics selects for the greatest bacterial load: the smile-frown transition. PLoS Biol 11: e1001540. doi: <u>10.1371/journal.pbio.1001540</u> PMID: <u>23630452</u>
- Hegreness M, Shoresh N, Damian D, Hartl D, Kishony R (2008) Accelerated evolution of resistance in multidrug environments. PNAS 105: 13977–13981. doi: 10.1073/pnas.0805965105 PMID: 18779569
- **15.** Li XZ, Nikaido H (2009) Efflux-mediated drug resistance in bacteria: an update. Drugs 69: 1555–1623. doi: <u>10.2165/11317030-00000000-00000</u> PMID: <u>19678712</u>
- Laehnemann D, Pena-Miller R, Rosenstiel P, Beardmore R, Jansen G, et al. (2014) Genomics of rapid adaptation to antibiotics: convergent evolution and scalable sequence amplification. Genome Biol Evol 6: 1287–1301. doi: <u>10.1093/gbe/evu106</u> PMID: <u>24850796</u>
- Gullberg E, Cao S, Berg OG, Ilbäck C, Sandegren L, et al. (2011) Selection of resistant bacteria at very low antibiotic concentrations. PLoS Pathog 7: e1002158. doi: <u>10.1371/journal.ppat.1002158</u> PMID: <u>21811410</u>
- Bosanquet AG, Bell PB (1996) Novel ex vivo analysis of nonclassical, pleiotropic drug resistance and collateral sensitivity induced by therapy provides a rationale for treatment strategies in chronic lymphocytic leukemia. Blood 87: 1962–1971. PMID: <u>8634445</u>
- Pluchino KM, Hall MD, Goldsborough AS, Callaghan R, Gottesman MM (2012) Collateral sensitivity as a strategy against cancer multidrug resistance. Drug Resist Updat 15: 98–105. doi: <u>10.1016/j.drup.</u> <u>2012.03.002</u> PMID: <u>22483810</u>
- Miles D, von Minckwitz G, Seidman AD (2002) Combination versus sequential single-agent therapy in metastatic breast cancer. Oncologist 7 Suppl 6: 13–19.
- Nishino K, Yamaguchi A (2001) Analysis of a complete library of putative drug transporter genes in Escherichia coli. J Bacteriol 183: 5803–5812. PMID: <u>11566977</u>
- 22. Fabrega A, Martin RG, Rosner JL, Tavio MM, Vila J (2010) Constitutive SoxS expression in a fluoroquinolone-resistant strain with a truncated SoxR protein and identification of a new member of the

marA-soxS-rob regulon, mdtG. Antimicrob Agents Chemother 54: 1218–1225. doi: <u>10.1128/AAC.</u> 00944-09 PMID: <u>20008776</u>

- Keren I, Shah D, Spoering A, Kaldalu N, Lewis K (2004) Specialized persister cells and the mechanism of multidrug tolerance in escherichia coli. J Bacteriol 186: 8172–8180. PMID: <u>15576765</u>
- Proctor RA, von Eiff C, Kahl BC, Becker K, McNamara P, et al. (2006) Small colony variants: a pathogenic form of bacteria that facilitates persistent and recurrent infections. Nat Rev Microbiol 4: 295–305. PMID: <u>16541137</u>
- Balke VL, Gralla JD (1987) Changes in the linking number of supercoiled dna accompany growth transitions in escherichia coli. J Bacteriol 169: 4499–4506. PMID: <u>3308843</u>
- Hill NS, Buske PJ, Shi Y, Levin PA (2013) A moonlighting enzyme links Escherichia coli cell size with central metabolism. PLoS Genet 9: e1003663. doi: <u>10.1371/journal.pgen.1003663</u> PMID: <u>23935518</u>
- Condon C, Liveris D, Squires C, Schwartz I, Squires CL (1995) rrna operon multiplicity in Escherichia coli and the physiological implications of rrn inactivation. J Bacteriol 177: 4152–4156. PMID: <u>7608093</u>
- Eichhorn E, van der Ploeg JR, Leisinger T (2000) Deletion analysis of the Escherichia coli taurine and alkanesulfonate transport systems. J Bacteriol 182: 2687–2695. PMID: <u>10781534</u>
- Greco WR, Bravo G, Parsons JC (1995) The search for synergy: a critical review from a response surface perspective. Pharmacol Rev 47: 331–385. PMID: <u>7568331</u>
- Lazar V, Pal Singh G, Spohn R, Nagy I, Horvath B, et al. (2013) Bacterial evolution of antibiotic hypersensitivity. Mol Syst Biol 9: 700. doi: <u>10.1038/msb.2013.57</u> PMID: <u>24169403</u>
- Szybalski W, Bryson V (1952) Genetic studies on microbial cross resistance to toxic agents. I. Cross resistance of Escherichia coli to fifteen antibiotics. J Bacteriol 64: 489–499. PMID: <u>12999676</u>
- Kaiser P, Regoes RR, Dolowschiak T, Wotzka SY, Lengefeld J, et al. (2014) Cecum lymph node dendritic cells harbor slow-growing bacteria phenotypically tolerant to antibiotic treatment. PLoS Biol 12: e1001793. doi: <u>10.1371/journal.pbio.1001793</u> PMID: <u>24558351</u>
- Clinical and Laboratory Standards Institute (2005) Performance Standards for Antimicrobial Susceptibility Testing. In: Fifteenth Informational Supplement. Wayne (Pennsylvania): Clinical and Laboratory Standards Institute.
- Kahlmeter G, Brown DFJ, Goldstein FW, MacGowan P, Mouton JW, et al. (2006) European Committee on Antimicrobial Susceptibility Testing (EUCAST) Technical Notes on antimicrobial susceptibility testing. Clin Microbiol Infect 12: 501–3. PMID: <u>16700696</u>
- Roberts JA, Kruger P, Paterson DL, Lipman J (2008) Antibiotic resistance—what's dosing got to do with it? Criti Care Med 36: 2433–40. doi: <u>10.1097/CCM.0b013e318180fe62</u> PMID: <u>18596628</u>
- 36. Ehrlich P (1913) Address in pathology on chemotherapeutics. The Lancet: 445-451.
- Fleming A (1964) Penicillin. Nobel Lectures, Physiology or Medicine 1942–1962. Elsevier Publishing Company.
- Drlica K (2003) The mutant selection window and antimicrobial resistance. J Antimicrob Chemother 52: 11–17. PMID: <u>12805267</u>
- Martinez MN, Papich MG, Drusano GL (2012) Dosing regimen matters: the importance of early intervention and rapid attainment of the pharmacokinetic/pharmacodynamic target. Antimicrob Agents Chemother 56: 2795–805. doi: <u>10.1128/AAC.05360-11</u> PMID: <u>22371890</u>
- 40. Subira M, Martino R, Gomez L, Marti JM, Estany C, et al. (2004) Low-dose amphotericin B lipid complex vs. conventional amphotericin B for empirical antifungal therapy of neutropenic fever in patients with hematologic malignancies—a randomized, controlled trial. Eur J Haematol 72: 342–347. PMID: 15059069
- PolyMedix (2013) Brilacidin (pmx-30063) antibiotic fact sheet. <u>http://goo.gl/8v3dct</u>. Accessed 17 February 2015.
- Huijben S, Bell AS, Sim DG, Tomasello D, Mideo N, et al. (2013) Aggressive chemotherapy and the selection of drug resistant pathogens. PLoS Pathog 9: e1003578. doi: <u>10.1371/journal.ppat.1003578</u> PMID: <u>24068922</u>
- Bergstrom CT, Lo M, Lipsitch M (2004) Ecological theory suggests that antimicrobial cycling will not reduce antimicrobial resistance in hospitals. PNAS 101: 13285–13290. PMID: <u>15308772</u>
- 44. Drusano GL, Fregeau C, Liu W, Brown DL, Louie A (2010) Impact of burden on granulocyte clearance of bacteria in a mouse thigh infection model. Antimicrob Agents Chemother 54: 4368–4372. doi: <u>10.</u> <u>1128/AAC.00133-10</u> PMID: <u>20516275</u>
- 45. Rex JH, Pfaller MA, Galgiani JN, Bartlett MS, Espinel-Ingroff A, et al. (1997) Development of interpretive breakpoints for antifungal susceptibility testing: conceptual framework and analysis of in vitro-in vivo correlation data for fluconazole, itraconazole, and candida infections. subcommittee on antifungal

susceptibility testing of the national committee for clinical laboratory standards. Clin Infect Dis 24: 235–247. PMID: <u>9114154</u>

- Louie A, Drusano GL, Banerjee P, Liu QF, Liu W, et al. (1998) Pharmacodynamics of fluconazole in a murine model of systemic candidiasis. Antimicrob Agents Chemother 42: 1105–1109. PMID: <u>9593135</u>
- 47. Jahn B, Martin E, Stueben A, Bhakdi S (1995) Susceptibility testing of candida albicans and aspergillus species by a simple microtiter menadione-augmented 3-(4,5-dimethyl-2-thiazolyl)-2,5-diphenyl-2h-tetrazolium bromide assay. J Clin Microbiol 33: 661–667. PMID: 7751374
- Greenberg ER, Anderson GL, Morgan DR, Torres J, Chey WD, et al. (2011) 14-day triple, 5-day concomitant, and 10-day sequential therapies for helicobacter pylori infection in seven latin american sites: a randomised trial. Lancet 378: 507–514. doi: <u>10.1016/S0140-6736(11)60825-8</u> PMID: <u>21777974</u>
- Goh EB, Yim G, Tsui W, McClure J, Surette MG, et al. (2002) Transcriptional modulation of bacterial gene expression by subinhibitory concentrations of antibiotics. Proceedings of the National Academy of Sciences 99: 17025–17030. PMID: 12482953
- Ohlsen K, Ziebuhr W, Koller KP, Hell W, Wichelhaus TA, et al. (1998) Effects of subinhibitory concentrations of antibiotics on alpha-toxin (hla) gene expression of methicillin-sensitive and methicillin-resistant Staphylococcus aureus isolates. Antimicrob Agents Chemother 42: 2817–2823. PMID: <u>9797209</u>
- Sakoulas G, Okumura C, Thienphrapa W, Olson J, Nonejuie P, et al. (2014) Nafcillin enhances innate immune-mediated killing of methicillin-resistant staphylococcus aureus. Journal of Molecular Medicine 92: 139–149. doi: <u>10.1007/s00109-013-1100-7</u> PMID: <u>24297496</u>
- Howden BP (2014) Antibiotics and *Staphylococcus aureus*—more than meets the MIC. J Mol Med (Berl) 92: 103–106. doi: <u>10.1007/s00109-013-1113-2</u> PMID: <u>24337505</u>
- Hilchie AL, Wuerth K, Hancock REW (2013) Immune modulation by multifaceted cationic host defense (antimicrobial) peptides. Nat Chem Biol 9: 761–768. doi: 10.1038/nchembio.1393 PMID: 24231617